Author’s response to reviews

Title: Hepatitis B Reactivation among 1962 Patients with Hematological Malignancy in Taiwan

Authors:

Chien-Yuan Chen (chiencyuanchen@ntu.edu.tw)
Feng-Ming Tien (b92401007@gmail.com)
Shang-Yi Huang (sy551225@ms7.hinet.net)
Wen-Chien Chou (whchou@ntu.edu.tw)
Ming Yao (yaoming@ntu.edu.tw)
Jih-Luh Tang (tangjh@ntu.edu.tw)
Hwei-Fang Tien (tienhf@ntu.edu.tw)
Wang-Huei Sheng (whsheng@ntu.edu.tw)
Aristine Cheng (arischeng@ntuh.gov.tw)

Version: 3 Date: 26 Nov 2017

Author’s response to reviews:

BMGE-D-16-00405R2

Hepatitis B Reactivation among 1962 Patients with Hematological Malignancy in Taiwan

Chien-Yuan Chen; Feng-Ming Tien; Shang-Yi Huang; Wen-Chien Chou; Ming Yao; Jih-Luh Tang; Hwei-Fang Tien; Wang-Huei Sheng; Aristine Cheng

BMC Gastroenterology

Dear editor:

Thank you for your kindly suggestion, we will revise as your suggestions.
1. Please use this form for the Availability of Data and Materials section: ‘The datasets generated and/or analysed during the current study are not publicly available in order to protect the participant’s identity but are available from the corresponding author on reasonable request.’

Reply: We do as editors’ suggestion.

2. Please state ‘not applicable’ in the Consent for Publication section; consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Reply: We do as editors’ suggestion.

3. Please state ‘The study received no funding’ in your Funding section.

Reply: We do as editors’ suggestion.

4. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Reply: We do as editors’ suggestion.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We do as editors’ suggestion.

Best wishes,

Wang-Huei Sheng

whsheng@ntu.edu.tw